Author's response to reviews

Title: Patient Engagement and Coaching for Health: The PEACH study - A cluster randomised controlled trial using the telephone to coach people with type 2 diabetes to engage with their GPs to improve diabetes care: A study protocol

Authors:

Doris Young (d.young@unimelb.edu.au)
John Furler (j.furler@unimelb.edu.au)
Margarite Vale (Margarite.vale@gmail.com)
Christine Walker (CWalker@chronicillness.org.au)
Leonie Segal (Leonie.segal@unisa.edu.au)
Patricia Dunning (trisha.dunning@barwonhealth.org.au)
James Best (jdbest@unimelb.edu.au)
Irene Blackberry (i.blackberry@unimelb.edu.au)
Ralph Audehm (ralph.audehm@rmh.org.au)
Nabil Sulaiman (nsulaiman@sharjah.ac.ae)
James Dunbar (Director@greaterhealth.org)
Patty Chondros (p.chondros@unimelb.edu.au)

Version: 3 Date: 11 April 2007

Author's response to reviews: see over
11 April 2007

BMC Editorial Production Team
BMC Family Practice

Dear Sir/Madam

Re: Submitted study protocol: Patient Engagement and Coaching for Health: The PEACH study – A cluster randomised controlled trial using the telephone to coach people with type 2 diabetes to engage with their GPs to improve diabetes care: A study protocol

MS 2897928821355319

Please find accompanying this letter our revised paper.

1. Title now changed to “A study protocol”

2. Authors names and order are exactly as in the online system

3. Affiliations are now in the required format and condensed to occupy only one page. Please note that some author’s affiliations are with non-government organisations and are thus written as “organisation, city, country”. If this is unacceptable please specify exactly how their affiliation ought to be written so that I can alter them to your requirements.

4. Abstract includes last section as “Trial Registration…” in the required format (This was always there and I have made no changes)

5. Word count now removed

6. Section headings now sentence case

7. Funding now within acknowledgements

8. Image files now contain no reference to title or figure number (now incorporated in MS file after references, with ‘Figure legends’ written above the legends, and the heading ‘Tables’ written above the tables. I will email the figure 2 to Tom Digby-Rogers as although it is a simple MS word figure on one page, cropped according to your instructions, it appears not to transfer to the web version satisfactorily.

I hope that this satisfies all the editorial requirements and look forward to publication of the manuscript.

Many thanks

Dr John Furler
Senior Lecturer
Department of General Practice
University of Melbourne
200 Berkeley St, Carlton, Victoria 3053
Australia
Phone  61 3 8344 4747
Fax  61 3 9347 6136
Email  j.furler@unimelb.edu.au